Author's response to reviews

Title: Development of an educational intervention for patients with Irritable Bowel Syndrome (IBS) - a pilot study.

Authors:

Gisela Ringstrom (gisela.ringstrom@vgregion.se)
Stine Storsrud (stine.storsrud@vgregion.se)
Sara Lundqvist (sara.lundqvist@vgregion.se)
Berndt Westman (berndt.westman@privat.utfors.se)
Magnus Simren (magnus.simren@medicine.gu.se)

Version: 3 Date: 28 January 2009

Author's response to reviews:

Regarding MS: 9268565921279939, entitled “Development of an educational intervention for patients with irritable bowel syndrome (IBS) - a pilot study?”

We would like to thank you for accepting our manuscript for publication in your journal. Corrections have been made according to the comments from your editorial production team, as well as your comments regarding the language.

Point-by-point response:

The manuscript has been copyedited by a native English speaking colleague, and some linguistic changes have been made.

Major revisions:

Table vertical elements are removed within the tables.
Figure titles are removed from the figure files.

Minor revision:

The header ‘Patient education in IBS, G Ringstom et al.’ is removed from all pages.
Short title and keywords are removed from the title page.
The section heading ‘ABSTRACT’ is changed to ‘Abstract’
All highlighting and tracking are removed from the manuscript.
The abbreviations are included as a paragraph instead of as a list.

On behalf of all the co-authors,